Author’s response to reviews

Title: An individually tailored family-centered intervention for pediatric obesity in primary care: Study protocol of a randomized type II hybrid implementation-effectiveness trial (Raising Healthy Children project)

Authors:

Justin Smith (jd.smith@northwestern.edu)
Cady Berkel (cady.berkel@asu.edu)
Neil Jordan (n-jordan@northwestern.edu)
David Atkins (datkins@uw.edu)
Shrikanth Narayanan (shri@sipi.usc.edu)
Carlos Gallo (carlos.gallo@northwestern.edu)
Kevin Grimm (kevin.j.grimm@asu.edu)
Thomas Dishion (dishion@asu.edu)
Anne Mauricio (anne.mauricio@asu.edu)
Jenna Rudo-Stern (jrudoste@asu.edu)
Mariah Meachum (mariah.meachum@northwestern.edu)
Emily Winslow (emily.winslow@asu.edu)
Meredith Bruening (meg.bruening@asu.edu)

Version: 1 Date: 06 Dec 2017

Author’s response to reviews:

Dear Editor,

Completed CONSORT Checklist now attached.